Author's response to reviews

Title: Acute spontaneous neck haematoma in infants: a rare entity

Authors:

    Shimin Zhuang (zhuangsm05@163.com)
    Jin Ye (yejin_sums@aliyun.com)
    Jingjia Li (li-jingjia@hotmail.com)

Version: 3
Date: 14 August 2014

Author's response to reviews:

Thanks for your kind advice in how to revise my manuscript, we have made the following changes in the revised manuscript:

1. We have removed the consent forms from the additional files;
2. ACKNOWLEDGEMENTS: people contributed towards the article have been listed as the authors, so there is not “acknowledgements section” in the article;
3. Ethics: we have declared the Consent as “Written informed consent was obtained from the patient for publication of this Case report and any accompanying images. A copy of the written consent is available for review by the Editor of this journal.”
4. LINE NUMBERING: We have revised our manuscript to include line and page numbers.